

## **Responses to reviewers**

**ESPS Manuscript NO: 15194**

**Title: Moxibustion Combining Acupuncture Increases Tight Junction Proteins' Expression in Crohn's Disease Patients**

Corrections on the manuscript were done:

1. Language certificate letter by professional English language editing companies were provided as an additional file. Minor revisions made by the English language polishing company were not highlighted in specific.
2. Title has been revised according to the 12-word limitation.
3. Running title has been added.
4. Files needed for Clinical Trials Study were provided, each in a separate PDF file, signed by Correspondence author.
5. All authors' abbreviation names were provided.
6. All headings' format has been revised according to the editor's comment.
7. Comments section was added as the editor requested.
8. The clean figures were separately provided in an additional excel.